

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2022GL101638

Key Points:

- Landward motion of the overriding plate, attributed to postseismic relaxation, was observed offshore 2 months after the Tohoku earthouake
- Models show that landward motion only transmits from the slab to the overriding plate, reaching the surface, if the megathrust is locked
- Observations imply shallow megathrust relocking within 2 months of the Tohoku earthquake, consistent with unstably sliding behavior

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. D'Acquisto, m.dacquisto@uu.nl

Citation:

D'Acquisto, M., & Govers, R. (2023). Offshore landward motion shortly after a subduction earthquake implies rapid relocking of the shallow megathrust. *Geophysical Research Letters*, 50, e2022GL101638. https://doi. org/10.1029/2022GL101638

Received 7 OCT 2022 Accepted 10 DEC 2022

Author Contributions:

Conceptualization: Rob Govers Data curation: Mario D'Acquisto Formal analysis: Mario D'Acquisto Funding acquisition: Rob Govers Investigation: Mario D'Acquisto Methodology: Mario D'Acquisto, Rob Govers Project Administration: Rob Govers Resources: Rob Govers Software: Rob Govers Supervision: Rob Govers Validation: Mario D'Acquisto, Rob Govers Visualization: Mario D'Acquisto

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Offshore Landward Motion Shortly After a Subduction Earthquake Implies Rapid Relocking of the Shallow Megathrust

Mario D'Acquisto¹ D and Rob Govers¹ D

¹Department of Earth Sciences, Faculty of Geoscience, Utrecht University, Utrecht, The Netherlands

Abstract Geodetic observations after large subduction earthquakes reflect multiple postseismic processes, including megathrust relocking. The timing of relocking and the observational constraints on it are unclear. Relocking was inferred to explain some observed landward motion that occurs within months. It was also considered unable to explain other, greater landward motion, including that off the coast of Japan beginning weeks after the 2011 Tohoku earthquake, attributed to postseismic relaxation. We use generic, 3D numerical models to show that relocking, particularly of the shallow interface, is needed for postseismic relaxation to produce landward motion on the tip of the overriding plate. We argue that this finding is consistent with previous simulations that implicitly relock the megathrust where afterslip is not included. We conclude that the Tohoku megathrust relocked within less than 2 months of the earthquake. This suggests that the shallow megathrust probably behaves as a true, unstably sliding asperity.

Plain Language Summary In the largest earthquakes, a tectonic plate suddenly slides under another, where previously the interface between them was largely locked. The deformation of the Earth's surface reflects multiple deep processes. One process is the restoration of interface locking. This means that elastic energy starts to accumulate again, building up toward the next earthquake. How long after an earthquake does the interface relock? Relocking is suggested to have caused some deformation observed months after some earthquakes, and so to have occurred by then. It is also thought to not explain the rapid landward motion of the seafloor on the upper plate, observed beginning less than 2 months after the 2011 Japan earthquake. We use numerical simulations to show that locking the shallow portion of the plate interface is needed for landward motion to transmit from the lower plate to the upper plate, as happened in Japan. Landward motion is understood to occur below and in the lower plate because of mantle flow and deep sliding taking place after the earthquake and caused by it. We conclude that this result agrees with simulations in previous studies, even though they were not interpreted in terms of locking by their authors.

1. Context, Aims, and Approach

The postseismic geodetic signal after a megathrust earthquake consists of several contributions: viscoelastic relaxation in the asthenospheric mantle, postseismic slip (afterslip) on the megathrust and its downdip continuation, poroelastic relaxation, and relocking of the megathrust interface (e.g., Bedford et al., 2016; Fialko, 2004; Hoffmann et al., 2018; Hu et al., 2014; Jónsson et al., 2003; Li et al., 2018, 2017; Peltzer et al., 1998; Remy et al., 2016; K. Wang et al., 2012). Figure 1 summarizes the main features of postseismic deformation at subduction zones. Relocking marks the beginning of the accumulation of slip deficit in the new earthquake cycle. It results in shortening of the overriding plate and near-trench landward velocities comparable to, but lower than, the plate convergence rate. As oceanward motion due to viscous relaxation and afterslip wane, the effect of the newly locked megathrust becomes apparent in geodetic displacement time series as landward motion of the surface of the overriding plate. Landward motion progressively reaches farther from the trench with time (K. Wang et al., 2012). Detecting the timing of relocking is critical for understanding the mechanical state of the megathrust system and properly assessing the associated seismic hazard.

Evidence suggests that relocking occurs rapidly (within weeks to months) after large megathrust earthquakes (Govers et al., 2018). In particular, rapid relocking has been inferred from the decomposition and inversion of Global Navigation and Satellite Systems (GNSS) displacement time series following the 2010 M_w 8.8 Maule (Chile) and 2007 M_w 8.0 Pisco (Peru) earthquakes (Bedford et al., 2016; Remy et al., 2016). Relocking was also inferred from the occurrence of a normal faulting intraplate earthquake 2 months after the 2006 Kuril Islands





Writing – original draft: Mario D'Acquisto, Rob Govers Writing – review & editing: Mario D'Acquisto, Rob Govers (Russia) megathrust rupture, followed 1 yr later by a thrust-fault intraplate earthquake (Lay et al., 2009), implying a rapid transition from extension to compression (Govers et al., 2018). Other evidence of the mechanical state of the megathrust may come from offshore observations, via Global Positioning System and acoustic (GPS-A) ranging, of landward horizontal postseismic motion close to the trench. However, these results have been inconclusive.

GPS-A observations at one offshore location above the 2005 M_w 7.2 Miyagi (Japan) megathrust earthquake indicate landward postseismic motion consistent with the signature of locking and interplate convergence starting 1 yr after the event (Sato et al., 2011). Landward motion was also detected at some offshore GPS-A sites on the overriding plate following the 2011 M_w 9.1 Tohoku (Japan) earthquake (Honsho et al., 2019; Sun et al., 2014; Tomita et al., 2017, 2014; Watanabe et al., 2014), beginning less than 2 months after the event (Japan Coast Guard, 2013). In contrast, the postseismic motion observed onshore was consistently oceanward over a period of at least 5 yr after the Tohoku earthquake (J. Wang et al., 2018). The offshore landward motion following the earthquake amounted to as much as 50 cm in the first year after the event (Sun et al., 2014). This amplitude is significantly more than can be explained merely by relocking and the far-field, steady-state convergence rate (8.3 cm/yr locally and as high as 9.1 cm/yr elsewhere on the Japan Trench; Watanabe et al., 2014). Watanabe et al. (2014) therefore explicitly concluded that relocking is irrelevant for explaining the observed motion after the Tohoku earthquake. Peña et al. (2019) reached the same conclusion for the onshore postseismic displacement field due to the Maule earthquake, whose observed trench-perpendicular components were entirely landward. Explanations have instead focused on postseismic relaxation, particularly viscoelastic relaxation and afterslip, since poroelastic relaxation can only account for relatively small signals (Hu et al., 2014; Peña et al., 2019).

Various studies interpreting the postseismic observations following the Tohoku earthquake agree that the offshore landward motion is primarily caused by viscous relaxation (Agata et al., 2019; Dhar et al., 2022; Freed et al., 2017; Fukuda & Johnson, 2021; Muto et al., 2019; Noda et al., 2018; Suito, 2017; Sun & Wang, 2015; Sun et al., 2014; Yamagiwa et al., 2015). Specifically, viscous flow in the sub-slab asthenosphere produces landward surface motion, while flow in the asthenospheric wedge produces oceanward motion (Suito, 2017). Muto et al. (2019) and Noda et al. (2018) also conclude that afterslip contributes to the offshore landward motion on the overriding plate. A consensus is thus forming that the typical evolution of the megathrust system outlined by K. Wang et al. (2012), in which landward motion appears progressively as the effect of locking prevails over diminishing postseismic relaxation, is somewhat complicated by sub-slab postseismic relaxation producing early and strong landward motion of the offshore forearc, regardless of locking. However, there has been no convincing mechanical explanation for why, and under what conditions, the landward motion resulting from viscous relaxation in the sub-slab mantle has a surface expression on the overriding plate.

We suspect that the postseismic observations of rapid landward motion, offshore on the overriding plate, require rapid relocking of the megathrust, as the latter determines the mechanical coupling between the two plates. We use a quasi-dynamic three-dimensional (3D) finite element method model with regularly repeating M_w 9.1 earthquakes to investigate this hypothesis (Figure 1 and Figure S1 in Supporting Information S1). We focus on large megathrust earthquakes that rupture the whole megathrust. We use a uniform slab profile cut perpendicularly across the Japan Trench from the Slab2 model (Hayes et al., 2018), but do not otherwise incorporate the structure nor aim to reproduce the specific observed deformation of the Japan subduction zone. We use uniform elastic plates and linear viscoelastic asthenospheric mantle with a viscosity of 10¹⁸ Pa-s in the postseismic period we study. We impose a constant motion at a rate of 90 mm/yr at either end of the downgoing plate. Slip deficit accumulates and is released according to megathrust locking and unlocking, similarly to Govers et al. (2018). We impose complete locking of portions of the megathrust (asperities) and allow frictionless creep on the rest of the megathrust and on its downdip continuation, consistent with observations (Hardebeck, 2015; Ikari et al., 2011; Scholz, 1998) and inverse model results (Herman & Govers, 2020). We use five asperities, rectangular in plan view, 50 km wide, centered 200 km along-trench from each other. We focus on the earthquake and postseismic deformation following the unlocking of the central asperity. This asperity extends horizontally from a distance of 133–3 km from the trench, reaching very close to it in agreement with the shallow coseismic slip observed during the Tohoku earthquake (e.g., Meng et al., 2011). The lateral asperities extend horizontally from 83 to 133 km from the trench and serve to provide megathrust coupling elsewhere along the plate margin, as observed along the Japan trench (Hashimoto et al., 2009; Johnson et al., 2016; Yamanaka & Kikuchi, 2004). To complete the supercycle, the two intermediate and two external lateral asperities are unlocked 20 and 40 yr after the central asperity, respectively. Every asperity is unlocked every 300 yr. When an asperity is unlocked, coseismic slip releases all





Figure 1. Schematic cross-section of deformation during the postseismic stage of the earthquake cycle. The subduction zone consists of two elastic lithospheric plates, separated by the megathrust interface and underlain by viscoelastic asthenospheric mantle. After the asperity ruptured during a megathrust earthquake, causing coseismically slip on the asperity and the rest of the megathrust, coseismic stress changes are relaxed. Postseismic relaxation includes bulk viscous flow in both asthenospheric domains, and rapid deep afterslip on the shear zone downdip of the megathrust. Postseismic relaxation produces additional convergent horizontal motion, with the oceanic domain moving landward and the continental/island arc domain moving landward. Landward surface motion is observed on land on the continental domain. After the 2011 Tohoku earthquake, opposing (landward) motion was observed on the seafloor at the tip of the overriding plate.

slip deficit, except for that due to locking of the other asperities. We treat the interface between slab and asthenospheric wedge, downdip of the megathrust, to be a viscoelastic shear zone (Tichelaar & Ruff, 1993; van Keken et al., 2002) with very low viscosity and the same elastic properties as the surrounding material (Govers et al., 2018; Muto et al., 2019). We let the shear zone accommodate relative motion at depths shallower than 80 km, where afterslip is commonly thought to occur (Diao et al., 2014; Freed et al., 2017; Hu et al., 2016; Sun et al., 2014; Yamagiwa et al., 2015). Afterslip occurs on the shear zone, immediately after the earthquake, relaxing as much as possible the elastic traction changes caused by coseismic deformation. We look at the cumulative horizontal surface motion during the first year after the earthquake, produced by the two major postseismic relaxation processes, afterslip and viscous relaxation, as well as by continued plate convergence. The modeling method is described more thoroughly in Text S1 in Supporting Information S1.

2. Model Results

In our first model, we immediately relock the entire ruptured asperity, from 3 km of horizontal distance from the trench down to a depth of 30 and 133 km horizontally from the trench. Figure 2 shows that the trench-perpendicular surface displacement in the first year after the earthquake is directed landward next to the trench on the overriding plate, as well as on the oceanic plate. Displacement is trenchward elsewhere on the overriding plate. The amplitude of landward displacement is largest (~30 cm) at the trench and decreases with distance from the trench and, to a lesser extent, with along-trench distance from the central asperity. Landward displacement reaches as far as ~45 km horizontally from the trench. Landward motion occurs throughout the oceanic domain, in the slab and sub-slab asthenosphere, and also extends

to the mechanically coupled forearc directly above the shallow megathrust, including the solid surface at the tip of the overriding plate (Figure S2 in Supporting Information S1).

In the second experiment, we do not relock any part of the central asperity after the earthquake, letting it slip freely postseismically together with the surrounding subduction interface. Displacement of the overriding plate is trenchward in this case, with no zone of landward motion close to the trench (Figures S3 and S4 in Supporting Information S1).

Does landward motion of the overriding plate require relocking of the entire asperity? Twenty kilometers was sometimes considered the updip limit of the frictionally unstable, potentially seismogenic portion of the megathrust (Byrne et al., 1988; Scholz, 1998). Relocking the deep part of the asperity only, between 20 and 30 km depth, is insufficient to produce landward displacement on the overriding plate during the first year after the earthquake (Figures S5 and S6 in Supporting Information S1). The amplitude of trenchward displacement above the asperity is nevertheless lower (i.e., less oceanward) than with no relocking whatsoever. We note that the shallow unlocked asperity and adjacent portions of the megathrust are not completely free to slip, because the mechanical continuity of both plates limits the slip that can occur in the vicinity of the relocked portion (Herman et al., 2018).

Relocking of only the shallow portion of the central asperity, above 20 km depth, yields landward postseismic surface displacement at the near-trench tip of the overriding plate, at the surface (Figure S7 in Supporting Information S1) and in the subsurface (Figure S8 in Supporting Information S1). However, the maximum surface amplitude of landward displacement in the first year (\sim 28 cm) and its trench-perpendicular spatial extent (\sim 25 km) are both smaller (slightly and substantially, respectively) than with relocking of the whole asperity.







Figure 2. The whole asperity relocks instantly after the earthquake. Cumulative horizontal postseismic surface displacement after 1 yr, assuming afterslip has ceased by then and including steady-state plate convergence. Colors show the trench-perpendicular displacement. Note that the color scale is asymmetrical, emphasizing the smaller (on the overriding plate) landward motion and de-emphasizing oceanward motion. Arrows show the displacement direction. The red rectangles show the surface projection of the outline of the asperities. The approximate location of the coastline, assumed to be directly above the downdip limit of the asperities is shown by the labeled line. The trench is shown as a barbed line. Dashed orange contours show coseismic displacement on the megathrust in 2.5 m increments. (a) Entire model, showing the surface motion in the central part of the model (marked and labeled in (a)). The extent of the relocked portion of the asperity, coinciding here with the whole asperity, is shown by the diagonal light green lines.

3. Discussion and Conclusions

3.1. Interpretation, Consistency With Previous Research, and Limitations

In our model results, the oceanic and continental domains generally have convergent, opposing motion, respectively landward and trenchward. Viscous flow and afterslip allow the shallow slab and the overriding plate in the vicinity of the rupture to further extend, continuing to release stresses accumulated during the interseismic period, as they accommodated convergence by shortening. This behavior is consistent with previously published results (e.g., Muto et al., 2019; Noda et al., 2018; Suito, 2017; Sun et al., 2014; Yamagiwa et al., 2015) and with the overview of deformation during megathrust earthquake cycles by K. Wang et al. (2012). Our results show that landward motion of the near-trench tip of the overriding plate requires mechanical coupling of the two plates on the megathrust. We interpret the cause of this for this dependency as follows: the near-trench portion of the slab, moving landward because of postseismic viscous relaxation and afterslip, transmits its landward motion to the thin portion of overriding plate directly above it, if and only if the locked megathrust provides mechanical coupling between them. Without any coupling, the megathrust slips freely and accommodates the opposite bulk motion of the two domains.

Our conclusion regarding the link between megathrust relocking and near-trench landward motion seemingly contradicts the studies that explain the postseismic displacement following the Tohoku earthquake with no need for locking (Agata et al., 2019; Dhar et al., 2022; Freed et al., 2017; Fukuda & Johnson, 2021; Muto et al., 2019; Noda et al., 2018; Suito, 2017; Sun & Wang, 2015; Sun et al., 2014; Yamagiwa et al., 2015), as well as the specific claim that relocking is largely irrelevant for explaining the observed offshore landward motion after the Tohoku earthquake (Watanabe et al., 2014). This apparent contradiction is due to those studies considering the effect of relocking to be only the signal due to continued far-field interplate convergence in addition to megathrust locking. This follows from the use of the back slip approach (Matsu'ura & Sato, 1989) in megathrust system models. However, when studying the early postseismic period, we must consider the effect of interplate coupling, if any, interacting with the motion due to postseismic relaxation, not merely with steady-state interplate convergence.

Previous studies generally consider afterslip to consist of all postseismic slip on the interface. This includes slip that releases slip deficit and associated elastic strains that were accumulated interseismically due to megathrust locking and not released coseismically. This afterslip is often assumed to decay in time while spatially stationary (e.g., Bedford et al., 2016). However, postseismic slip also occurs passively on unlocked portions of the megathrust as a result of interplate convergence and viscous relaxation. Since a lack of afterslip implies that the interface is locked, that is, does not slip, we can interpret the results of previous studies in terms of relocking by observing the afterslip distributions they use. Suito (2017) does not include afterslip in his model. Sun et al. (2014) impose an afterslip distribution that has near-zero values in the entirety of the rupture zone. Diao et al. (2014) and Noda et al. (2018) invert exclusively onshore displacement time series into afterslip, with different slab geometries. They find no significant afterslip above a depth of 20 km below solid ground, corresponding to ~150 and \sim 90 km of horizontal distance from the trench, respectively, in each geometry. Freed et al. (2017) and Yamagiwa et al. (2015) invert onshore and offshore geodetic displacement time series into afterslip. They find little afterslip above 25 km depth below sea level, or within ~150 km of horizontal distance from the trench, in the region where offshore landward displacements are observed, at any time after the earthquake, although there is some shallow afterslip immediately to the south and, in the results of Freed et al., farther north. Dhar et al. (2022) and Muto et al. (2019) invert onshore and offshore geodetic observations with stress-driven afterslip. Muto et al. (2019) use a two-dimensional (2D) model and find no afterslip within 130 km of horizontal distance from the trench and only very minor afterslip between 130 and 150 km. Dhar et al. (2022) use a 3D model and find a lack of afterslip as deep as 40 km below sea level in the central part of the rupture, and generally shallower than 30 km except for shallow afterslip to the south of the coseismic rupture zone and of the observed postseismic landward motion. In all these models, then, the shallow interface is implicitly locked postseismically at the location where landward postseismic displacement is observed geodetically and produced by the models themselves.

The peak amplitude of landward displacement in our fully relocked model results (30 cm) is smaller than that (50 cm) observed in the first year after the Tohoku earthquake and reproduced by previous modeling studies (e.g., Sun et al., 2014). The reason for the smaller amplitude in our results is possibly due to the relatively limited peak amplitude of coseismic slip (27 m). Our choice of a single, uniform Maxwell viscosity also probably plays a role. In fact, previous studies find that explaining observations requires a spatially inhomogeneous, temporally variable effective viscosity, with lower short-term values particularly at the base of the slab (Agata et al., 2019, 2018;

Freed et al., 2017; Muto et al., 2019; Sun et al., 2014). However, there is no widespread agreement and best-fitting viscosities vary greatly with modeling methodology (Fukuda & Johnson, 2021). We do not aim to specifically reproduce observations and thus we do not explore different amplitudes and spatio-temporal distributions of viscosity that previous studies consider. Other models study megathrust behavior and include sophisticated methodologies that go beyond continuum mechanics and can provide insight into the specifics of the relocking mechanism. For instance, Caniven et al. (2021) use a discrete element method (Cundall & Strack, 1979; Morgan, 2015) that considers elastic particles interacting frictionally and indicates that megathrust relocking is controlled by fault dilatancy and contraction. Our models do not approach the full physical complexity of megathrust and bulk deformation at subduction zones. Rather, they merely establish that relocking, whatever its cause, is needed for the landward motion due to postseismic relaxation to extend from the slab to the top of the overriding plate, consistently with the results of previous modeling studies. Future work that considers additional complexity can investigate the role of specific relaxation and relocking mechanisms and patterns in explaining specific GNSS observations. Additionally, some interpretations of the likely behavior of the megathrust in vicinity of the Tohoku rupture can be made from the need of relocking to explain the observed presence of landward motion of the offshore portion of the overriding plate.

3.2. Spatio-Temporal Features of Relocking and Fault Friction

The necessity of a locked megathrust implies that relocking has already occurred when landward motion is observed there. In the case of the Tohoku earthquake, the earliest, less reliable postseismic GPS-A observations indicate that landward motion was already occurring less than 2 months after the event (Japan Coast Guard, 2013). This implies that relocking occurred within a few weeks of the Tohoku earthquake. Rapid relocking was previously inferred from the decomposition of geodetic displacement time series after other large megathrust earthquakes, specifically the 2010 M_w 8.8 Maule (Chile) and 2007 M_w 8.0 Pisco (Peru) earthquakes (Bedford et al., 2016; Remy et al., 2016). Dixon et al. (2014), Malservisi et al. (2015), and Voss et al. (2017) invert GNSS time series into afterslip and slow-slip events (SSEs) after the much smaller 2012 M_w 7.6 Nicoya earthquake. They find that afterslip decays with a dominant characteristic timescale of 70 days (Malservisi et al., 2015), after which SSEs only occur later and outside of the coseismic rupture area. Rapid relocking is thus not without support in previous research analyzing geodetic observations.

From the perspective of fault friction theories, rapid relocking of a ruptured fault asperity is compatible with the rate-and-state formulation (Dieterich, 1979; Ruina, 1983). In fact, if the asperity is understood as an unstably sliding, rate-weakening portion of the fault, near-instantaneous relocking is expected, as it immediately follows the end of the unstable sliding episode that constitutes the earthquake. According to the common understanding of rate-and-state fault friction and its role in earthquake cycle behavior, afterslip and stable creep (i.e., a lack of interseismic coupling) occur outside of the velocity-weakening, unstably sliding, seismogenic portion of the fault. This behavior can be seen, for instance, in the synoptic model of Marone (1998), the rupture simulation of Barbot et al. (2012) for part of the San Andreas fault, and the sophisticated numerical model of postseismic relaxation after the Tohoku earthquake of Muto et al. (2019).

Our model results indicate that near-trench landward postseismic motion of the overriding plate is not produced by relocking the asperity at intermediate depths (between 20 and 30 km below sea level), at which the Tohoku earthquake rupture is thought to have nucleated (Chu et al., 2011; Freed et al., 2017; Japan Meteorological Agency, 2012). Instead, relocking the uppermost portion (above 20 km depth) is necessary and sufficient for producing landward postseismic motion on the overriding plate, although with a smaller amplitude and spatial extent than when the deeper asperity is also relocked. This finding is based on our rather coarse examination, considering cumulatively all postseismic deformation during the first year after the earthquake and only one depth limit between the shallow and deeper megathrust. However, it is consistent with the results of more sophisticated simulations (Freed et al., 2017; Muto et al., 2019; Noda et al., 2018; Sun & Wang, 2015; Sun et al., 2014; Yamagiwa et al., 2015), when interpreting afterslip in terms of locking or lack thereof. Our conclusions indicate that offshore geodetic observations can detect the spatio-temporal evolution of megathrust relocking, which is relevant for seismic hazard assessment and provides insights into the nature of the shallow megathrust. Our work can thus be seen as a pilot study for the SZ4D MegaArray initiative (McGuire et al., 2017).

Our results show a need for rapid relocking of the shallow megathrust to produce landward postseismic motion offshore on the overriding plate. At the same time, the rapidly relocked area of a fault is generally understood to

be a frictionally unstable interface. The most straightforward interpretation of these two factors together suggests that the shallow megathrust in the Tohoku region is a true asperity: unstably sliding, slipping coseismically and frictionally locked during the interseismic stage. This is in contrast with the conventional expectation that the shallow megathrust be stably sliding and unlocked interseismically (Loveless & Meade, 2010; Tsuru et al., 2000). This wisdom was partly challenged by observations of unusually large slip hosted by the shallow megathrust during the Tohoku earthquake (Meng et al., 2011). Herman et al. (2018) pointed out that slip deficit can accumulate on a low-friction portion of the megathrust passively locked by the mechanical continuity of the plates and the adjacent truly locked asperity, in a process they called pseudo-coupling. Noda and Lapusta (2013) instead proposed a shallow megathrust with frictional properties that allow it to be stably sliding interseismically and still slip substantially coseismically in a dynamic rupture simulation. These proposed mechanisms could reconcile stably sliding interseismic behavior with large coseismic slip. Furthermore, it is unclear whether the behavior of the Tohoku megathrust in the most recent earthquake reflects permanent frictional properties of the megathrust, or whether these properties changed during the last cycle.

In light of our model results, the rapid postseismic landward motion observed offshore the Tohoku earthquake suggests that the shallow megathrust there behaved as a true asperity, at least during the last cycle, without requiring the rupture to extend beyond the unstably sliding portion. This would imply that the nature of the shallow megathrust has been misunderstood and in fact resembles that of the deeper megathrust, despite the less consolidated material and different ambient conditions at shallower depths. Alternatively, the shallow megathrust could be stably sliding and have only hosted substantial coseismic slip because of dynamic rupture effects and/ or pseudo-coupling. In this case, an explanation must be provided to explain sufficient coupling of the shallow megathrust shortly after the earthquake as to produce postseismic landward motion. In general, future research should use tailored numerical models to test what degree of locking is compatible with the offshore postseismic observations, considering different spatial and temporal distributions of afterslip together with a realistic geometry and rheology.

The conclusion that rapid shallow megathrust relocking is needed to explain the offshore postseismic GPS-A observations following the Tohoku earthquake is in agreement with similar conclusions of rapid relocking drawn from onshore GNSS observations for other earthquakes, such as the 2010 Maule and 2007 Pisco earthquakes (Bedford et al., 2016; Remy et al., 2016). This suggests that rapid relocking is not restricted to a specific event or subduction zone. Near-trench, postseismic geodetic observations should detect evidence of shallow megathrust relocking, if any, following earthquakes producing substantial postseismic relaxation. However, a lack of postseismic landward motion, as for the first year following the 2005 Miyagi earthquake (Sato et al., 2011), does not necessarily imply a general lack of relocking. In particular, the lack of landward motion might be explained by locking being restricted to greater depths (implying along-trench variations in the frictional character of the shallow megathrust). Alternatively, it might be due to the oceanward signal of deep afterslip overcoming the small landward signal due to interplate convergence in the absence of substantial viscous relaxation, as for the metation of afterslip distribution with the specific local geometry might determine whether the limit of the landward signature of deep afterslip occurs on the overriding plate, as proposed by Noda et al. (2018).

Megathrust relocking, including at shallow depths, is needed to allow postseismic relaxation, which produce landward motion in the slab and sub-slab mantle, to also affect the near-trench tip of the overriding plate. Offshore postseismic observations can differentiate between a locked and unlocked shallow megathrust and provide evidence of very rapid (within 2 months) shallow relocking after the Tohoku event. Observations suggest that the shallow megathrust above 20 km depth in the Tohoku region behaves as an unstably sliding asperity.

Data Availability Statement

The model output files that we used for the figures of this paper are digitally stored in the Yoda repository of Utrecht University and are available under the CC-BY license at https://doi.org/10.24416/UU01-SS41UK.



Acknowledgments

This work was partly funded by Dutch Research Council (NWO) Grant ALWGO.2017.007. The mesh generator program Gmsh (Geuzaine & Remacle, 2009) was used to make the finite element meshes for the numerical models. The Generic Mapping Tools (Wessel et al., 2019) and Adobe Illustrator were used for visualization.

References

- Agata, R., Barbot, S. D., Fujita, K., Hyodo, M., Iinuma, T., Nakata, R., et al. (2019). Rapid mantle flow with power law creep explains deformation after the 2011 Tohoku mega-quake. *Nature Communications*, 10(1), 1–11. https://doi.org/10.1038/s41467-019-08984-7
- Agata, R., Ichimura, T., Hori, T., Hirahara, K., Hashimoto, C., & Hori, M. (2018). An adjoint-based simultaneous estimation method of the asthenosphere's viscosity and afterslip using a fast and scalable finite-element adjoint solver. *Geophysical Journal International*, 213(1), 461–474. https://doi.org/10.1093/gji/ggx561
- Barbot, S., Lapusta, N., & Avouac, J.-P. (2012). Under the hood of the earthquake machine: Toward predictive modeling of the seismic cycle. *Science*, 336(6082), 707–710. https://doi.org/10.1126/science.1218796
- Bedford, J., Moreno, M., Li, S., Oncken, O., Baez, J. C., Bevis, M., et al. (2016). Separating rapid relocking, afterslip, and viscoelastic relaxation: An application of the postseismic straightening method to the Maule 2010 cGPS. *Journal of Geophysical Research: Solid Earth*, *121*(10), 7618–7638. https://doi.org/10.1002/2016JB013093
- Byrne, D. E., Davis, D. M., & Sykes, L. R. (1988). Loci and maximum size of thrust earthquakes and the mechanics of the shallow region of subduction zones. *Tectonics*, 7(4), 833–857. https://doi.org/10.1029/TC007i004p00833
- Caniven, Y., Morgan, J. K., & Blank, D. G. (2021). The role of along-fault dilatancy in fault slip behavior. *Journal of Geophysical Research: Solid Earth*, *126*(11), e2021JB022310. https://doi.org/10.1029/2021JB022310
- Chu, R., Wei, S., Helmberger, D. V., Zhan, Z., Zhu, L., & Kanamori, H. (2011). Initiation of the great Mw 9.0 Tohoku-oki earthquake. Earth and Planetary Science Letters, 308(3), 277–283. https://doi.org/10.1016/j.epsl.2011.06.031
- Cundall, P. A., & Strack, O. D. L. (1979). A discrete numerical model for granular assemblies. *Géotechnique*, 29(1), 47–65. https://doi.org/10.1680/geot.1979.29.1.47
- Dhar, S., Muto, J., Ito, Y., Miura, S., Moore, J. D. P., Ohta, Y., & Iinuma, T. (2022). Along-arc heterogeneous rheology inferred from post-seismic deformation of the 2011 Tohoku-oki earthquake. *Geophysical Journal International*, 230(1), 202–215. https://doi.org/10.1093/gjj/ggac063
- Diao, F., Xiong, X., Wang, R., Zheng, Y., Walter, T. R., Weng, H., & Li, J. (2014). Overlapping post-seismic deformation processes: Afterslip and viscoelastic relaxation following the 2011 Mw 9.0 Tohoku (Japan) earthquake. *Geophysical Journal International*, 196(1), 218–229. https:// doi.org/10.1093/gji/ggt376
- Dieterich, J. H. (1979). Modeling of rock friction: 1. Experimental results and constitutive equations. Journal of Geophysical Research: Solid Earth, 84(B5), 2161–2168. https://doi.org/10.1029/JB084iB05p02161
- Dixon, T. H., Jiang, Y., Malservisi, R., McCaffrey, R., Voss, N., Protti, M., & Gonzalez, V. (2014). Earthquake and tsunami forecasts: Relation of slow slip events to subsequent earthquake rupture. *Proceedings of the National Academy of Sciences*, 111(48), 17039–17044. https://doi. org/10.1073/pnas.1412299111
- Fialko, Y. (2004). Evidence of fluid-filled upper crust from observations of postseismic deformation due to the 1992 Mw 7.3 Landers earthquake. Journal of Geophysical Research: Solid Earth, 109(B8). https://doi.org/10.1029/2004JB002985
- Freed, A. M., Hashima, A., Becker, T. W., Okaya, D. A., Sato, H., & Hatanaka, Y. (2017). Resolving depth-dependent subduction zone viscosity and afterslip from postseismic displacements following the 2011 Tohoku-oki, Japan earthquake. *Earth and Planetary Science Letters*, 459, 279–290. https://doi.org/10.1016/j.epsl.2016.11.040
- Fukuda, J., & Johnson, K. M. (2021). Bayesian inversion for a stress-driven model of afterslip and viscoelastic relaxation: Method and application to postseismic deformation following the 2011 Mw 9.0 Tohoku-oki earthquake. *Journal of Geophysical Research: Solid Earth*, 126(5), e2020JB021620. https://doi.org/10.1029/2020JB021620
- Geuzaine, C., & Remacle, J.-F. (2009). Gmsh: A 3-D finite element mesh generator with built-in pre- and post-processing facilities. *International Journal for Numerical Methods in Engineering*, 79(11), 1309–1331. https://doi.org/10.1002/nme.2579
- Govers, R., Furlong, K. P., van de Wiel, L., Herman, M. W., & Broerse, T. (2018). The geodetic signature of the earthquake cycle at subduction zones: Model constraints on the deep processes. *Reviews of Geophysics*, 56(1), 6–49. https://doi.org/10.1002/2017RG000586
- Hardebeck, J. L. (2015). Stress orientations in subduction zones and the strength of subduction megathrust faults. *Science*, 349(6253), 1213–1216. https://doi.org/10.1126/science.aac5625
- Hashimoto, C., Noda, A., Sagiya, T., & Matsu'ura, M. (2009). Interplate seismogenic zones along the Kuril-Japan trench inferred from GPS data inversion. *Nature Geoscience*, 2(2), 141–144. https://doi.org/10.1038/ngeo421
- Hayes, G. P., Moore, G. L., Portner, D. E., Hearne, M., Flamme, H., Furtney, M., & Smoczyk, G. M. (2018). Slab2, a comprehensive subduction zone geometry model. *Science*, 362(6410), 58–61. https://doi.org/10.1126/science.aat4723
- Herman, M. W., Furlong, K. P., & Govers, R. (2018). The accumulation of slip deficit in subduction zones in the absence of mechanical coupling: Implications for the behavior of megathrust earthquakes. *Journal of Geophysical Research: Solid Earth*, 123(9), 8260–8278. https://doi. org/10.1029/2018JB016336
- Herman, M. W., & Govers, R. (2020). Locating fully locked asperities along the South America subduction megathrust: A new physical interseismic inversion approach in a Bayesian framework. *Geochemistry, Geophysics, Geosystems*, 21(8). https://doi.org/10.1029/2020GC009063
- Hoffmann, F., Metzger, S., Moreno, M., Deng, Z., Sippl, C., Ortega-Culaciati, F., & Oncken, O. (2018). Characterizing afterslip and ground displacement rate increase following the 2014 Iquique-Pisagua Mw 8.1 earthquake, Northern Chile. *Journal of Geophysical Research: Solid Earth*, 123(5), 4171–4192. https://doi.org/10.1002/2017JB014970
- Honsho, C., Kido, M., Tomita, F., & Uchida, N. (2019). Offshore postseismic deformation of the 2011 Tohoku earthquake revisited: Application of an improved GPS-acoustic positioning method considering horizontal gradient of sound speed structure. *Journal of Geophysical Research: Solid Earth*, 124(6), 5990–6009. https://doi.org/10.1029/2018JB017135
- Hu, Y., Bürgmann, R., Freymueller, J. T., Banerjee, P., & Wang, K. (2014). Contributions of poroelastic rebound and a weak volcanic arc to the postseismic deformation of the 2011 Tohoku earthquake. *Earth Planets and Space*, 66(1), 106. https://doi.org/10.1186/1880-5981-66-106
- Hu, Y., Bürgmann, R., Uchida, N., Banerjee, P., & Freymueller, J. T. (2016). Stress-driven relaxation of heterogeneous upper mantle and time-dependent afterslip following the 2011 Tohoku earthquake. *Journal of Geophysical Research: Solid Earth*, 121(1), 385–411. https://doi. org/10.1002/2015JB012508
- Ikari, M. J., Marone, C., & Saffer, D. M. (2011). On the relation between fault strength and frictional stability. Geology, 39(1), 83–86. https:// doi.org/10.1130/G31416.1
- Japan Coast Guard. (2013). Seafloor movements observed by seafloor geodetic observations after the 2011 off the Pacific coast of Tohoku Earthquake. Retrieved from https://cais.gsi.go.jp/YOCHIREN/report/kaihou89/03_04.pdf
- Japan Meteorological Agency. (2012). Information on the 2011 Great East Japan Earthquake. Retrieved from https://www.jma.go.jp/jma/ en/2011_Earthquake/Information_on_2011_Earthquake.html
- Johnson, K. M., Mavrommatis, A., & Segall, P. (2016). Small interseismic asperities and widespread aseismic creep on the northern Japan subduction interface. *Geophysical Research Letters*, 43(1), 135–143. https://doi.org/10.1002/2015GL066707

19448007, 2023,

1, Downlo

- Jónsson, S., Segall, P., Pedersen, R., & Björnsson, G. (2003). Post-earthquake ground movements correlated to pore-pressure transients. *Nature*, 424(6945), 179–183. https://doi.org/10.1038/nature01776
- Lay, T., Kanamori, H., Ammon, C. J., Hutko, A. R., Furlong, K. P., & Rivera, L. (2009). The 2006–2007 Kuril Islands great earthquake sequence. Journal of Geophysical Research: Solid Earth, 114(B11). https://doi.org/10.1029/2008JB006280
- Li, S., Bedford, J., Moreno, M., Barnhart, W. D., Rosenau, M., & Oncken, O. (2018). Spatiotemporal variation of mantle viscosity and the presence of cratonic mantle inferred from 8 yr of postseismic deformation following the 2010 Maule, Chile, earthquake. *Geochemistry, Geophys*ics, Geosystems, 19(9), 3272–3285. https://doi.org/10.1029/2018GC007645
- Li, S., Moreno, M., Bedford, J., Rosenau, M., Heidbach, O., Melnick, D., & Oncken, O. (2017). Postseismic uplift of the Andes following the 2010 Maule earthquake: Implications for mantle rheology. *Geophysical Research Letters*, 44(4), 1768–1776. https://doi.org/10.1002/2016GL071995
- Loveless, J. P., & Meade, B. J. (2010). Geodetic imaging of plate motions, slip rates, and partitioning of deformation in Japan. Journal of Geophysical Research: Solid Earth, 115(B2), B02410. https://doi.org/10.1029/2008JB006248
- Malservisi, R., Schwartz, S. Y., Voss, N., Protti, M., Gonzalez, V., Dixon, T. H., et al. (2015). Multiscale postseismic behavior on a megathrust: The 2012 Nicoya earthquake, Costa Rica. Geochemistry, Geophysics, Geosystems, 16(6), 1848–1864. https://doi.org/10.1002/2015GC005794
- Marone, C. (1998). Laboratory-derived friction laws and their application to seismic faulting. *Annual Review of Earth and Planetary Sciences*, 26(1), 643–696. https://doi.org/10.1146/annurev.earth.26.1.643
- Matsu'ura, M., & Sato, T. (1989). A dislocation model for the earthquake cycle at convergent plate boundaries. *Geophysical Journal Interna*tional, 96(1), 23–32. https://doi.org/10.1111/j.1365-246X.1989.tb05247.x
- McGuire, J. J., Plank, T., Barrientos, S. E., Becker, T. W., Brodsky, E., Cottrell, E., et al. (2017). The SZ4D initiative: Understanding the processes that Underlie subduction zone hazards in 4D. Vision document submitted to the National Science Foundation. The IRIS Consortium. Retrieved from https://www.sz4d.org/_files/ugd/66466d_c5202b9573e1413eb006995304a4b274.pdf
- Meng, L., Inbal, A., & Ampuero, J.-P. (2011). A window into the complexity of the dynamic rupture of the 2011 Mw 9 Tohoku-oki earthquake. Geophysical Research Letters, 38(7). https://doi.org/10.1029/2011GL048118
- Morgan, J. K. (2015). Effects of cohesion on the structural and mechanical evolution of fold and thrust belts and contractional wedges: Discrete element simulations. *Journal of Geophysical Research: Solid Earth*, 120(5), 3870–3896. https://doi.org/10.1002/2014JB011455
- Muto, J., Moore, J. D. P., Barbot, S., Iinuma, T., Ohta, Y., & Iwamori, H. (2019). Coupled afterslip and transient mantle flow after the 2011 Tohoku earthquake. *Science Advances*, 5(9), eaaw1164. https://doi.org/10.1126/sciadv.aaw1164
- Noda, A., Takahama, T., Kawasato, T., & Matsu'ura, M. (2018). Interpretation of offshore crustal movements following the 2011 Tohoku-oki earthquake by the combined effect of afterslip and viscoelastic stress relaxation. *Pure and Applied Geophysics*, *175*(2), 559–572. https://doi.org/10.1007/s00024-017-1682-z
- Noda, H., & Lapusta, N. (2013). Stable creeping fault segments can become destructive as a result of dynamic weakening. *Nature*, 493(7433), 518–521. https://doi.org/10.1038/nature11703
- Peltzer, G., Rosen, P., Rogez, F., & Hudnut, K. (1998). Poroelastic rebound along the Landers 1992 earthquake surface rupture. Journal of Geophysical Research: Solid Earth, 103(B12), 30131–30145. https://doi.org/10.1029/98JB02302
- Peña, C., Heidbach, O., Moreno, M., Bedford, J., Ziegler, M., Tassara, A., & Oncken, O. (2019). Role of lower crust in the postseismic deformation of the 2010 Maule earthquake: Insights from a model with power law rheology. *Pure and Applied Geophysics*, 176(9), 3913–3928. https:// doi.org/10.1007/s00024-018-02090-3
- Remy, D., Perfettini, H., Cotte, N., Avouac, J. P., Chlieh, M., Bondoux, F., et al. (2016). Postseismic relocking of the subduction megathrust following the 2007 Pisco, Peru, earthquake. Journal of Geophysical Research: Solid Earth, 121(5), 3978–3995. https://doi.org/10.1002/2015JB012417
- Ruina, A. (1983). Slip instability and state variable friction laws. Journal of Geophysical Research: Solid Earth, 88(B12), 10359–10370. https://doi.org/10.1029/JB088iB12p10359
- Sato, M., Saito, H., Ishikawa, T., Matsumoto, Y., Fujita, M., Mochizuki, M., & Asada, A. (2011). Restoration of interplate locking after the 2005 Off-Miyagi Prefecture earthquake, detected by GPS/acoustic seafloor geodetic observation. *Geophysical Research Letters*, 38(1). https://doi. org/10.1029/2010GL045689
- Scholz, C. H. (1998). Earthquakes and friction laws. Nature, 391(6662), 37-42. https://doi.org/10.1038/34097
- Suito, H. (2017). Importance of rheological heterogeneity for interpreting viscoelastic relaxation caused by the 2011 Tohoku-Oki earthquake. *Earth, Planets, and Space, 69*(1), 21. https://doi.org/10.1186/s40623-017-0611-9
- Sun, T., & Wang, K. (2015). Viscoelastic relaxation following subduction earthquakes and its effects on afterslip determination. Journal of Geophysical Research: Solid Earth, 120(2), 1329–1344. https://doi.org/10.1002/2014JB011707
- Sun, T., Wang, K., Iinuma, T., Hino, R., He, J., Fujimoto, H., et al. (2014). Prevalence of viscoelastic relaxation after the 2011 Tohoku-oki earthquake. *Nature*, 514(7520), 84–87. https://doi.org/10.1038/nature13778
- Tichelaar, B. W., & Ruff, L. J. (1993). Depth of seismic coupling along subduction zones. *Journal of Geophysical Research: Solid Earth*, 98(B2), 2017–2037. https://doi.org/10.1029/92JB02045
- Tomita, F., Kido, M., Ohta, Y., Iinuma, T., & Hino, R. (2017). Along-trench variation in seafloor displacements after the 2011 Tohoku earthquake. *Science Advances*, 3(7), e1700113. https://doi.org/10.1126/sciadv.1700113
- Tomita, F., Kido, M., Osada, Y., Azuma, R., Hino, R., Fujimoto, H., et al. (2014). Postseismic seafloor deformation associated with the 2011 Tohoku-oki earthquake revealed by repeated GPS/acoustic observations. Oral presentation presented at the International Symposium on Geodesy for Earthquake and Natural Hazards (GENAH), Matsushima, Japan.
- Tsuru, T., Park, J.-O., Takahashi, N., Kodaira, S., Kido, Y., Kaneda, Y., & Kono, Y. (2000). Tectonic features of the Japan Trench convergent margin off Sanriku, northeastern Japan, revealed by multichannel seismic reflection data. *Journal of Geophysical Research: Solid Earth*, 105(B7), 16403–16413. https://doi.org/10.1029/2000JB900132
- van Keken, P. E., Kiefer, B., & Peacock, S. M. (2002). High-resolution models of subduction zones: Implications for mineral dehydration reactions and the transport of water into the deep mantle. *Geochemistry, Geophysics, Geosystems*, 3(10). https://doi.org/10.1029/2001GC000256
- Voss, N. K., Malservisi, R., Dixon, T. H., & Protti, M. (2017). Slow slip events in the early part of the earthquake cycle. Journal of Geophysical Research: Solid Earth, 122(8), 6773–6786. https://doi.org/10.1002/2016JB013741
- Wang, J., Mann, P., & Stewart, R. R. (2018). Late Holocene structural style and seismicity of highly transpressional faults in southern Haiti. *Tectonics*, 37(10), 3834–3852. https://doi.org/10.1029/2017TC004920
- Wang, K., Hu, Y., & He, J. (2012). Deformation cycles of subduction earthquakes in a viscoelastic Earth. Nature, 484(7394), 327–332. https:// doi.org/10.1038/nature11032
- Watanabe, S., Sato, M., Fujita, M., Ishikawa, T., Yokota, Y., Ujihara, N., & Asada, A. (2014). Evidence of viscoelastic deformation following the 2011 Tohoku-Oki earthquake revealed from seafloor geodetic observation. *Geophysical Research Letters*, 41(16), 5789–5796. https://doi.org/10.1002/2014GL061134

- Wessel, P., Luis, J. F., Uieda, L., Scharroo, R., Wobbe, F., Smith, W. H. F., & Tian, D. (2019). The generic mapping tools version 6. Geochemistry, Geophysics, Geosystems, 20(11), 5556–5564. https://doi.org/10.1029/2019GC008515
- Yamagiwa, S., Miyazaki, S., Hirahara, K., & Fukahata, Y. (2015). Afterslip and viscoelastic relaxation following the 2011 Tohoku-oki earthquake (Mw 9.0) inferred from inland GPS and seafloor GPS/Acoustic data. *Geophysical Research Letters*, 42(1), 66–73. https://doi.org/10.1002/2014GL061735
- Yamanaka, Y., & Kikuchi, M. (2004). Asperity map along the subduction zone in northeastern Japan inferred from regional seismic data. Journal of Geophysical Research: Solid Earth, 109(B7). https://doi.org/10.1029/2003JB002683

References From the Supporting Information

- Almeida, R., Lindsey, E. O., Bradley, K., Hubbard, J., Mallick, R., & Hill, E. M. (2018). Can the updip limit of frictional locking on megathrusts be detected geodetically? Quantifying the effect of stress shadows on near-trench coupling. *Geophysical Research Letters*, 45(10), 4754–4763. https://doi.org/10.1029/2018GL077785
- Balay, S., Abhyankar, S., Adams, M. F., Benson, S., Brown, J., Brune, P., et al. (2021a). PETSc web page. 30 April 2022 Retrieved from https://petsc.org/
- Balay, S., Abhyankar, S., Adams, M. F., Benson, S., Brown, J., Brune, P., et al. (2021b). *PETSc/TAO users manual (No. ANL-21/39-Revision 3.16)*. Argonne National Laboratory.
- Balay, S., Gropp, W. D., McInnes, L. C., & Smith, B. F. (1997). Efficient management of parallelism in object oriented numerical software libraries. In E. Arge, A. M. Bruaset, & H. P. Langtangen (Eds.), Modern software tools in scientific computing (pp. 163–202). Birkhäuser Press.
- Cambiotti, G. (2020). Joint estimate of the coseismic 2011 Tohoku earthquake fault slip and post-seismic viscoelastic relaxation by GRACE data inversion. *Geophysical Journal International*, 220(2), 1012–1022. https://doi.org/10.1093/gji/ggz485
- Fletcher, R. (1988). Practical methods of optimization. Wiley.
- Gabriel, E., Fagg, G. E., Bosilca, G., Angskun, T., Dongarra, J. J., Squyres, J. M., et al. (2004). Open MPI: Goals, concept, and design of a next generation MPI implementation. In D. Kranzlmüller, P. Kacsuk, & J. Dongarra (Eds.), *Recent advances in parallel virtual machine and message passing interface* (Vol. 3241, pp. 97–104). Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-540-30218-6_19 CUPCCO Constitution for an environment of the second sec
- GEBCO Compilation Group. (2022). GEBCO_2022 grid. https://doi.org/10.5285/e0f0bb80-ab44-2739-e053-6c86abc0289c
- Hetland, E. A., & Hager, B. H. (2006). Interseismic strain accumulation: Spin-up, cycle invariance, and irregular rupture sequences. *Geochemistry, Geophysics, Geosystems*, 7(5). https://doi.org/10.1029/2005GC001087
- Kawakatsu, H., Kumar, P., Takei, Y., Shinohara, M., Kanazawa, T., Araki, E., & Suyehiro, K. (2009). Seismic evidence for sharp lithosphere-asthenosphere boundaries of oceanic plates. *Science*, 324(5926), 499–502. https://doi.org/10.1126/science.1169499
- Kumar, P., & Kawakatsu, H. (2011). Imaging the seismic lithosphere-asthenosphere boundary of the oceanic plate. *Geochemistry, Geophysics, Geosystems*, 12(1). https://doi.org/10.1029/2010GC003358
- Melosh, H. J., & Williams, C. A. (1989). Mechanics of graben formation in crustal rocks: A finite element analysis. Journal of Geophysical Research: Solid Earth, 94(B10), 13961–13973. https://doi.org/10.1029/JB094iB10p13961
- Verfürth, R. (1994). A posteriori error estimation and adaptive mesh-refinement techniques. Journal of Computational and Applied Mathematics, 50(1), 67–83. https://doi.org/10.1016/0377-0427(94)90290-9
- Wang, K., Zhu, Y., Nissen, E., & Shen, Z.-K. (2021). On the relevance of geodetic deformation rates to earthquake potential. *Geophysical Research Letters*, 48(11), e2021GL093231. https://doi.org/10.1029/2021GL093231
- Zhu, Y., Wang, K., & He, J. (2020). Effects of earthquake recurrence on localization of interseismic deformation around locked strike-slip faults. Journal of Geophysical Research: Solid Earth, 125(8). https://doi.org/10.1029/2020JB019817